

Author's response to reviews

Title:Immigrant-Native Differences in Caries-related Knowledge, Attitude, and Oral Health Behaviors: a Cross-sectional Study in Taiwan

Authors:

Chih-Chang Chen (ccchen@mail.iosh.gov.tw)
Shang-Jyh Chiou (chiouatg@gmail.com)
Chun-Chan Ting (ccting@kmu.edu.tw)
Ying-Chun Lin (bonnie0925.tw@yahoo.com.tw)
Chih-Cheng Hsu (cch@nhri.org.tw)
Fu-Li Chen (ph1007@mails.fju.edu.tw)
Chien-Hung Lee (cnhung@kmu.edu.tw)
Ted Chen (tchen@tulane.edu)
Ya-Ying Lin (yayinglin@yahoo.com)
Chin-Shun Chang (csc1201@seed.net.tw)
Hsiao-Ling Huang (hhuang@kmu.edu.tw)

Version:4**Date:**7 January 2014

Author's response to reviews: see over

Jan 6, 2014

Dear Editor-in-Chief,

The authors have included sections and answered the questions as following.

Competing interests

Please include a 'Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests.'

We added the 'Competing interests' section. (Please see P. 20)

The following five questions that are asked of authors are:

-In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.

NO

-Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.

NO.

-Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.

NO.

-Do you have any other financial competing interests? If so, please specify.

Non-financial competing interests: are there any non-financial competing interests (political, personal, religious, academic, ideological, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.

NO.

Authors' contributions

-Please include an 'Authors' contributions' section before the Acknowledgements and Reference list.

We have included Authors' contributions section in P. 20-21

Please let us know if you have any further concerns.

Yours Sincerely,

Hsiao-Ling Huang, Dr.PH

Associate Professor

Department of Oral Hygiene, College of Dental Medicine

Kaohsiung Medical University, Taiwan

100, Shih-Chuan 1st Road, Kaohsiung, 80708, Taiwan.

E-mail: hhuang@kmu.edu.tw

Phone number: +011-886-7-3121101 ext. 2159

Fax number: +011-886-7-3235767